Multi-Object Auctions with Package Bidding: An Experimental Comparison of Vickrey and *i*BEA*

Yan Chen Kan Takeuchi

May 1, 2008

^{*}We thank Ted Bergstrom, Vincent Crawford, Tom Finholt, Ted Groves, Dan Houser, Elena Katok, Nancy Kotzian, Tony Kwasnica, John Ledyard, Dan Levin, Kevin McCabe, Lydia Mechtenberg, David Parkes, David Reiley, Tatsuyoshi Saijo, Tim Salmon and seminar participants at Meinheim, Michigan, Penn State, Shanghai Jiaotong, UC Santa Barbara, UC San Diego, the 2004 Economic Science Association Meetings (Amsterdam and Tucson), and the Eighth Experimental Economics Conference at Osaka University (November 2004) for helpful discussions, and Alexandra Achen for excellent research assistance. We are grateful to two anonymous referees and the Associate Editor for helpful comments and suggestions, which significantly improved the paper. Chen gratefully acknowledges financial support from the National Science Foundation through grant no. SES-0339587. Chen: School of Information, University of Michigan, 1075 Beal Avenue, Ann Arbor, MI 48109-2112. Email: yanchen@umich.edu. Takeuchi: Division of the Humanities and Social Sciences, California Institute of Technology, 1200 East California Blvd., MC 228-77, Pasadena, CA 91125. Email: ktakeuch@umich.edu.

Abstract

The use of package auctions for complex resource allocation has been rapidly increasing in recent years. In this paper, we study two package auction mechanisms in a laboratory setting, a sealed bid Vickrey auction and an ascending version of Vickrey, the *i*BEA auction. Unlike the single-unit Vickrey auction, where bidders tend to overbid in the laboratory, most of our bidders either underbid or bid their true values. Furthermore, at the aggregate level, while the *i*BEA auction generates significantly higher bidder profit, the Vickrey auction generates significantly higher revenue than does *i*BEA. While the two auctions achieve the same efficiency, a higher proportion of *i*BEA auctions achieves 100% efficiency than does the Vickrey auction. We also find that human bidders learn from their robot opponents when the robot strategies are (myopic) best responses.

Keywords: combinatorial auctions, procurement auctions, experimental economics **JEL Classifications**: C91, D44

1 Introduction

In recent years, the use of various multi-object auction mechanisms to determine resource allocation has been rapidly increasing. In particular, the Federal Communications Commission (FCC) spectrum auctions, characterized by synergies across licenses, stimulated tremendous research interests in complex auction design for multiple objects and with synergies. Since 1994, the FCC has used the Simultaneous Multiple Round (SMR) auction to allocate spectrum licenses and raised over \$41 billion in revenue (Milgrom 2004). However, this auction format does not allow package bidding. Ledyard, Porter and Rangel (1997) demonstrate that the performance of the FCC design degrades in the presence of complementarities. More generally, when bidder valuations for multiple objects are super-additive, package bidding is necessary to increase efficiency, seller revenue and bidder willingness to participate (Bykowsky, Cull and Ledyard 2000). Economic research on package auction mechanisms helps policy makers such as the FCC as well as others who must allocate complex resources. For example, based on experimental tests of various package auction mechanisms by Brunner, Goeree, Holt and Ledyard (2006) and Goeree and Holt (2007), the FCC has decided to use the hierarchical package bidding auction mechanism proposed by Rothkopf, Pekec and Harstad (1998) for selling spectrum licenses in the 700 MHz band.¹

The earliest research on package auction design is a proposal for a sealed bid combinatorial auction sale of paired airport takeoff and landing slots proposed by Rassenti, Smith and Bulfin (1982). In another study, Banks, Ledyard and Porter (1989) present two kinds of iterative package auctions to allocate uncertain and unresponsive resources, AUSM (Adaptive User Selection Mechanism) and the iterative Vickrey-Clarke-Groves mechanism (Vickrey (1961), Clarke (1971), Groves (1973)). In laboratory experiments, these package auctions significantly outperform markets and administrative procedures. More recently, Kwasnica, Ledyard, Porter and DeMartini (2005) create and test a new design for multiobject iterative auctions by merging the better features of the AUSM and the FCC SMR designs. The resulting new Resource Allocation Design (RAD) is shown to perform better than either parent.

Another important application of package auction design is in a Business-to-Business

¹See http://wireless.fcc.gov/auctions/default.htm?job=auction_summary&id=73 for more information.

(B2B) context, which is predominantly multi-object and often involves synergies. Package auctions have the potential to provide value to both buyers and sellers of goods and services. The descending Dutch auction used to sell flowers in Aalsmeer, Holland is a simple version of a package auction for homogeneous goods (Katok and Roth 2004). More recently, package auctions have been successfully applied to transportation procurement and sourcing networks. For example, Sears Logistics Services is the first procurer of trucking services to use a package auction to reduce its costs. Since 1993, it consistently saved 13 percent over past procurement practices, mostly through bilateral negotiations (Ledyard, Olson, Porter, Swanson and Torma 2002). Similarly, Proctor & Gamble has successfully used package auction in its sourcing of displays and truckload transportation services, with estimated savings of 9.6 percent in its two and a half years of implementation (Sandholm, Levine, Concordia, Martyn, Hughes, Jacobs and Begg 2006). Finally, the London bus routes provide an example of the use of a package auction format in public procurement. In this instance, the local transportation authority has adopted a form of package auction because of expected economic synergies among routes located in the same area of London. The London bus routes auction has led to increased quality of service and lower costs, and thus is considered a success (Cantillon and Pesendorfer 2006).

Empirical investigations of package auctions have important implications not only in procurement and privatization, but also potentially in the allocation of scarce equipment time in large scientific collaboratories.² The National Science Foundation and other federal agencies are poised to make significant investments in expanding the ability of geographicallydistributed groups of scientists to conduct research via the Internet. There are currently over eighty collaboratories in use within multiple scientific communities, including space physics, HIV/AIDS, software engineering and neuroimaging (Finholt 2002). In many collaboratories, a critical feature of the equipment time allocation problem is that contiguous time slots are more valuable than the sum of separate slots, i.e., user valuation for multiple slots exhibits synergy. Therefore, package auctions might be an important mechanism in achieving efficient allocation of equipment time (Takeuchi, Lin, Chen and Finholt 2008).

Despite successful results for package auctions, the theoretical properties of these auc-

²First proposed in the late eighties, a collaboratory is a center without walls, in which researchers can perform their research without regard to physical location - interacting with colleagues, accessing instrumentation, sharing data and computational resources, and accessing information in digital libraries (Wulf 1993).

tions are not often understood. Therefore, it is unclear whether a given choice of design option is the most appropriate one.

An important standard for nearly all mechanism design work, and for auctions in particular, is the Vickrey-Clarke-Groves (VCG) mechanism. The VCG mechanism is dominant strategy incentive compatible, i.e., bidding one's true valuation is always optimal regardless of others' strategies. Furthermore, it implements the efficient outcome. For the single object case, VCG mechanism becomes the familiar second-price auction. In the auction context, we follow convention and call the VCG mechanism Vickrey auctions.

However, despite its attractive theoretical properties, the Vickrey auction has some disadvantages. In the package auction context, there are three main concerns.³ First, the Vickrey auction might be vulnerable to collusion. For example, bidders have the incentive to use shill bidders through which they could manipulate the allocation and prices in their favor. Second, it might suffer from the monotonicity problem,⁴ i.e., adding bidders might reduce equilibrium revenues. Third, previous laboratory experiments show that, in the single object case, the dominant strategy in second-price auctions is not transparent. Many experimental subjects consistently overbid in second-price auctions and do not seem to learn from prior experimence (Kagel 1995).

Contrary to the economist belief that Vickrey auctions are rarely used in practice, Lucking-Reiley (2000b) presents evidence that Vickrey auctions have long been the predominant auction format for mail sales of collectible postage stamps, at least 65 years earlier than the publication of Vickrey's seminal paper. Vickrey-like auctions have also been appearing in auctions on the Internet, sometimes with an additional feature of "proxy bidding." In these auctions, a bidder tells his proxy his maximum willingness to pay. The proxy keeps this information secret and bids on the bidder's behalf in an ascending auction in a pre-announced increment. If every bidder uses a proxy, then the bidder with the highest maximum price wins and pays (approximately) the second highest price. The most prominent examples of such auctions include Amazon and eBay.⁵

To retain the advantages of the sealed-bid Vickrey auctions while reducing their disad-

³Note the first two problems do not appear when all goods are substitutes for all bidders.

⁴See Milgrom (2004) Chapter 8 for examples.

⁵Ockenfels and Roth (2002) analyze the closing rules of eBay and Amazon. They show that eBay auctions, with a hard closing rule, give bidders incentives for sniping, while Amazon auctions, with a soft closing rule, do not give such incentives and hence are a more faithful dynamic version of Vickrey.

vantages, researchers have searched for an ascending bid package auction with comparable theoretical properties. A recent example is the iBundle Extend & Adjust (*i*BEA) auction (Mishra and Parkes 2007), which is built on the iBundle auction of Parkes and Ungar (2000). It is an ascending bid auction with package bidding which implements efficient allocation and Vickrey payments in *ex post* Nash equilibrium, with only a free-disposal requirement on agent preferences. The *i*BEA auction is the first ascending package Vickrey auction for general valuations with a single price path.⁶ Thus, theoretically, *i*BEA represents a major advance in modeling package auctions.

In this paper, we investigate two package auction mechanisms in the laboratory, Vickrey and *i*BEA, to evaluate their performance among boundedly rational individuals. Our study advances research on auctions in two ways. First, compared to previous experimental studies of package auctions, the theoretical properties for these two auctions are well understood. Second, this is the first experimental study of *i*BEA, a new promising ascending bid package auction mechanism.

We study these two mechanisms in a simple environment where three bidders compete for four items (and thus 15 packages), with synergies across subsets of the items. In our human-robot treatments, each human bidder competes against two automated bidders. In some treatments, the automated bidders are programmed to follow the dominant strategy in Vickrey, and Myopic Best Response in *i*BEA, while in the other treatments, automated bidders are programmed to follow random strategies. The use of automated agents serve two purposes. First, it allows the experimenter to compare the performance of the two mechanisms in an environment free from the strategic uncertainties inherent in interactions between human bidders.⁷ Second, the use of automated agents is becoming increasingly widespread in Internet auctions, which allow for conveniently asynchronous bidding (Lucking-Reiley 2000a). Therefore, it is important to study how humans react when they bid against automated agents. For robustness check, we implement two humanhuman treatments, where human bidders compete against each other in the same economic environment as in the human-robot treatments.

The main results of our two auction scenarios present some unexpected findings. Unlike

 $^{^{6}}$ The *i*BEA auction with a proxy behaves equivalently to the ascending proxy auction (Ausubel and Milgrom 2002) when buyers are substitutes (Mishra and Parkes 2007).

⁷See, e.g., Kagel and Levin (2001) for an experiment with human bidders against automated agents in multi-unit auctions of homogeneous goods.

the single-unit Vickrey auction, where bidders tend to overbid in the laboratory, most of our bidders either underbid or bid their true values. Furthermore, at the aggregate level, while the *i*BEA auction generates significantly higher bidder profit, the Vickrey auction generates significantly higher revenue. While the two auctions achieve the same efficiency, a higher proportion of *i*BEA auctions achieves 100% efficiency than does the Vickrey auction. We also find that human bidders learn from their robot opponents when the robot strategies are (myopic) best responses.

The rest of the paper is organized as follows. Section 2 introduces the auction mechanisms. Section 3 presents the experimental design. Section 4 presents the hypotheses. Section 5 presents the analysis and main results. Section 6 concludes.

2 The Auctions

In this section, we introduce our two auction mechanisms. To do so, we first set up a simple framework that allows us to explain the auctions clearly.

In this framework, let $N = \{1, ..., n\}$ be a finite set of bidders. Let *i* denote an agent, where i = 0 is an *auctioneer* and i > 0 is a bidder. $N_0 = N \cup \{0\}$ is the set of all bidders and the auctioneer. Let $K = \{1, ..., k\}$ represent the set of objects to be sold, and $X = \{0, 1\}^k$ represent the set of combinations of objects. Let \mathcal{B}_i be a set of *bids* of bidder *i*. Each bid is a pair, (\mathbf{x}, p) , where $\mathbf{x} \in X$ corresponds to the packages desired and $p \in \mathbb{R}$ is the bid price. Let $v_i : X \to \mathbb{R}_+$ be bidder *i*'s valuation function that assigns a value to a package. Finally, $\delta = (\delta_1, \ldots, \delta_b)$ is an indicator vector, where $\delta_j \in \{0, 1\}$ indicates whether bid (x_j, p_j) is winning or losing, and where *b* is the total number of bids.

As the winner determination problem is part of every package auction design, we formally define this problem for our experiment.

Definition (Package auction winner determination problem). The package auction winner determination problem is to maximize the sum of all bids, indicating each bid as *winning* or *losing*, under the constraint that each item can be sold to at most one bidder:

$$\max_{\delta} \sum_{j=1}^{b} \delta_j p_j \quad \text{subject to} \quad \sum_{j \in \{j:\delta_j=1\}} x_j \le (1, 1, \dots, 1).$$
(1)

One of the most important goals in multi-object auctions is for all bidders to bid truthfully. We now define a truthful bid: a bid $(\mathbf{x}, p) \in \mathcal{B}_i$ is a *truthful bid*, if $v_i(\mathbf{x}) = p$. The allocation associated with the solution to the winner determination problem maximizes the aggregate surplus, when every bidder bids for all packages and the bids are all truthful.

Although it is desirable that all bidders truthfully bid for all packages, in theory, depending on the auction mechanism, it might not always be optimal for a bidder to bid her true value, and in practice, it is not easy for bidders to calculate all values for all possible packages. To design a good auction mechanism, we would like to obtain the most efficient allocation among feasible outcomes. In the design of a multi-object auction mechanism, there are three potential problems to overcome.

- 1. The exposure problem: When items are not substitutes and bidders cannot bid on packages, bidders are usually exposed to the risk that they may overpay. For example, suppose that there are two items and that bidder *i*'s valuations exhibit strong complementarity, such that $v_i((1,0)) = v_i((0,1)) = 1 < 3 = v_i((1,1))$. He may bid on each item at prices more than 1, expecting that he gets both of the items. However, it is possible that he could get only one item while paying more than 1. Auctions with package bidding, such as the two auctions in this study, should overcome this problem.
- 2. The threshold problem: Suppose that there are four bidders and three items to trade and that v₁((1,0,0)) = v₂((0,1,0)) = v₃((0,0,1)) = 1.5 and v₄((1,1,1)) = 4. In this setting, it is efficient to allocate the items to Bidders 1, 2 and 3. However, suppose that Bidders 1, 2 and 3 bid on their desirable item at price 1, respectively, and that Bidder 4 bids on the package of all items at price 3.6. In this case, the winning bid is Bidder 4's bid, but none of Bidders 1-3 can overbid on Bidder 4's bid. The ascending bid auction, *i*BEA, overcomes the threshold problem, which we will explain in Section 2.2.
- 3. Computational problems: There are two computational problems in package auctions, the auctioneer's and the bidders'. For the auctioneer, the problem is to solve Eq. (1), which is NP-complete (Rothkopf et al. 1998). For the bidders, they must evaluate all possible combinations, which can be cognitively difficult. Evaluation of the latter is where experimental research can be especially valuable.

In the following sections, we first introduce the Vickrey and *i*BEA auctions, and then discuss their respective theoretical properties.

2.1 Vickrey Auction with Package Bidding

A Vickrey auction with package bidding is an extension of the more familiar second-price auction. At the beginning of each auction, each bidder selects the packages he would like to bid on, and the amount he would like to bid for each package. Each bidder can choose to bid on as many packages as he wants, and he can bid on a single object multiple times, by bidding on several packages that contain that item. However, no matter how many packages a bidder bids on, he will never win more than one package. This type of bidding is called an exclusive-or (XOR) bid. Since all items are weakly complements in our experiment, XOR is not necessary for incentive compatibility. However, to minimize the interface difference between the two mechanisms, we impose XOR bids in the Vickrey auction, as they are required in the iBEA auction.

Next, once all bidders have submitted their bids, the auctioneer will choose the combination of submitted bids that yields the highest sum of bids. The set of bidders winning a package are the winning bidders.

After determining the winning bidders, the auctioneer then, one at a time, chooses each winning bidder as a **pivotal bidder**. The auctioneer examines the bids again, but ignores the bids of the pivotal bidder. The auctioneer determines the allocation of goods that maximizes the sum of bids, using the same rules as before, but not considering any bids placed by the pivotal bidder. Once this new allocation has been determined, the auctioneer compares the sum of bids generated by this allocation with those generated when no bids are excluded.

At the end of the auction, the amount that the winning bidders are required to pay depends on the additional revenue that each bidder generated, which is calculated by comparing the original revenue obtained by the auctioneer versus the revenue obtained by the auctioneer when the given bidder is pivotal. The following example from the experimental instruction illustrates how this process works. The fictitious currency used in this example (and throughout the experiment) is pounds (\pounds) .

Example: Suppose that there are three bidders and four objects to allocate, and the following bids are submitted:

	Package	Price	Status
Bidder 1	AB	£50	winning
Bidder 2	$\mathbf{C}\mathbf{D}$	£40	winning
Bidder 3	ABCD	£60	
Bidder 3	AB	£30	

As shown in the table, the bids from Bidder 1 and 2 are the winning bids, because they generate the highest revenue for the auctioneer $\pounds 50 + \pounds 40 = \pounds 90$.

However, the auctioneer does not ask Bidder 1 to pay $\pounds 50$. Suppose we choose Bidder 1 as a pivotal bidder, and ignore his bids. The winning bids then become Bidder 2's bid on **CD** and Bidder 3's bid on **AB**.

	Package	Price	Status
Bidder 1	AB-	£50	winning
Bidder 2*	CD	£40	winning
Bidder 3	ABCD	£60	
Bidder 3*	AB	£30	winning

In this case, the auctioneer calculates the revenue that those winning bids would generate, which is $\pounds 40 + \pounds 30 = \pounds 70$. Thus, the additional revenue that Bidder 1 generates is $\pounds 20$, since $\pounds 90 - \pounds 70 = \pounds 20$. This $\pounds 20$ is the price adjustment for Bidder 1. Therefore, Bidder 1 pays $\pounds 50$ and receives $\pounds 20$ back. His final price is $\pounds 30$.

The Vickrey auction is dominant strategy incentive compatible and implements an efficient outcome. Each bidder ends up with a Vickrey payoff in equilibrium. However, it has some shortcomings. First, it is vulnerable to collusion. Second, the revenue under a Vickrey auction can be very low. Third, the dominant strategy in a Vickrey auction might not be transparent when it is implemented with boundedly rational people. Many previous experimental studies on single unit auctions (see Kagel (1995) for a survey) demonstrate that bidders systematically overbid in single-unit Vickrey auctions. More recently, Isaac and James (2000) show that, in an experimental setting with two items (and thus three packages), Vickrey auctions with package bidding consistently generate higher efficiency than those without package bidding. However, they do not compare a Vickrey package auction with an ascending package auction.

2.2 *i*Bundle Extend & Adjust (*i*BEA) Auction

The *i*Bundle Extend & Adjust (*i*BEA) auction is proposed by Mishra and Parkes (2007), which builds on the iBundle auction described in Parkes and Ungar (2000). The former does not retain several details in the latter, such as the ε -best response set and the ε last-andfinal bid, which are adopted in this experiment. The *i*BEA auction is an ascending-price generalized Vickrey auction. It maintains non-linear and non-anonymous prices on packages, and terminates with approximately efficient allocation and Vickrey payments. To achieve these properties, the mechanism requires a myopic best response, which is an *ex post* Nash equilibrium. In an *i*BEA auction, each auction takes place in several rounds. Let ε be the price increment. The choice of ε involves a tradeoff between the speed of the auction and the closeness to efficiency of the final outcome. That is, a smaller ε can achieve more efficient outcomes, at the cost of a longer auction. An outline of the *i*BEA process is as follows:

- 1. The auctioneer initializes prices for all packages.
- 2. At the beginning of each round, for each bidder, the auctioneer announces "ask prices" $\mathbf{p}_i(t)$ for all packages to bidder *i*.
- 3. Given the prices, each bidder can bid on as many packages as she wants, and can bid on a single item multiple times, by bidding on several packages that contain that item. Each bidder's submitted bids, $\mathcal{B}_i(t)$, must satisfy the following rules:
 - (a) Winning bid resubmission rule: $\exists (\mathbf{x}, p) \in \mathcal{B}_i(t)$, such that (\mathbf{x}, p) is a winning bid in the previous round. That is, if a bidder has made a winning bid in the previous round, she is obligated to bid on that package, at the same price, in the next round. Once a bid is losing, a bidder has no further commitment to bid on that package, unless he chooses to do so.
 - (b) Last-and-final bid: The last-and-final option allows a bidder to continue to bid for a package when the bid price is narrowly above the object's value. Once a bidder chooses the last-and-final option on a package, he receives a small discount, ε, and the last-and-final bid is automatically resubmitted at the same price in every round until the auction terminates. Therefore, even if the price for that package increases, a bidder cannot increase his bid on that package.

The last-and-final option facilitates the computation of Vickrey prices. It also reveals to the auctioneer a bidder's approximate true value for a package.

- 4. If there are no new bids, then the auction terminates. Otherwise, given $\{\mathcal{B}_i(t)\}_{i \in N}$, the auctioneer solves Eq. (1) and revises ask prices to each bidder in the following manner:
 - (a) The ask prices for a bidder with any winning bid(s) remain the same as in the previous round.
 - (b) The prices for packages with last-and-final bids remain the same as in the previous round.
 - (c) The price for each package of a losing bid increases by ε .
 - (d) All ask prices are adjusted to be self-consistent, i.e., the price for a package should not be less than the price for any of its subsets.

The auctioneer proceeds to the next round t + 1, then returns to step 2.

In the *i*BEA auction, there are two phases to determine the auction's outcome. The auctioneer uses Phase I to determine the final allocation and Phase II to compute the final prices and Vickrey discounts. Phase I terminates when all agents who submit bids are assigned a package, and the allocation is the final allocation. The auction then proceeds to Phase II. In each round of Phase II, the auctioneer selects a pivotal bidder and ignores this bidder's bids to compute his *externality*. When she computes all externalities, the prices are determined, from which she computes the Vickrey payoffs. Note that all adjustments to Vickrey payoffs made in *i*BEA are with respect to ask prices and the marginal effect on efficiency.

Just as Ausubel and Milgrom (2002) rely on the assumption of a straightforward bidding strategy, in an *i*BEA auction, it is assumed that each bidder takes a *Myopic Best-Response* (MBR). We say bidder *i* takes a myopic best-response to the ask prices $\mathbf{p}_i(t)$ announced by the auctioneer, if

$$\mathcal{B}_{i}(t) = \left\{ (x,p) \middle| v_{i}(x,p) - p \ge \max\left\{ \max_{z \in X} \left\{ v_{i}(z,p(z)) - p(z) \right\}, 0 \right\} - \varepsilon \text{ and } p = p(x) \right\}.$$
(2)

where p(x) is the ask price of package x for bidder i. The MBR strategy chooses ε -maximized packages, i.e., packages arbitrarily close to the best package.

We now explain MBR in more detail. We define a bidder's temporary profit as $v_i(x, p(x)) - p(x)$. Therefore, when a bidder examines his menu, he first looks for any packages on which he has a negative temporary profit. For those packages that have a negative profit, he adds ε to his temporary profit, because he knows that he can buy the package with an ε discount using the last-and-final option. He does not change his temporary profit for a package on which he currently has a positive profit. He then examines the revised temporary profits for all packages, and finds the package that gives him the greatest revised temporary profit. There are two possibilities:

- 1) If his greatest possible profit is greater than or equal to ε , he will bid on all packages that give him a revised temporary profit within ε of the maximum temporary profit. For example, if one package gives him a temporary profit of £17, no package gives him a profit of more than £17, and $\varepsilon = \pounds 5$, then he will bid on all packages that give him a temporary profit of at least £17 – £5, or £12.
- 2) If his greatest possible profit is less than ε , he will bid only on those packages that have a revised temporary profit greater than or equal to 0.

Under the assumption of a MBR, we can achieve competitive equilibrium prices and efficiency. More precisely, if bidders follow the MBR, then the vector of ask prices $\{\mathbf{p}_i(t)\}_{i\in N}$ is a competitive equilibrium price vector after the end of Phase I. As $\varepsilon \to 0$, the final allocation becomes arbitrarily close to the efficient allocation. Furthermore, Phase II allows the auctioneer to compute Vickrey payoffs. Parkes and Ungar (2000) prove that the MBR is incentive compatible: MBR is an *ex post* Nash equilibrium of *i*BEA, as $\varepsilon \to 0$.

To illustrate how the *i*BEA auction works, we use a simple three-bidder, two-item example. In this example, we assume that all bidders follow a MBR strategy. Furthermore, we set $\varepsilon = 5$, as in our experimental setting.

[Table 1 about here.]

Table 1 illustrates how *i*BEA works. The top panel presents Phases I and II round-byround and the bottom panel compares the results with those of the Vickrey auction.

As shown in Table 1, for Bidder 1, the values of \mathbf{A} , \mathbf{B} and \mathbf{AB} are (10, 0, 10). For Bidders 2 and 3, they are (0, 30, 30) and (4, 14, 22), respectively.

In round 1, the offered prices are the same. The auctioneer breaks the tie by randomly choosing as many winning bidders as possible, and Bidders 1 and 2 are selected. Since

Bidder 3 does not win any package, the auctioneer raises the offered prices for Bidder 3. Specifically, the price for the losing bid on **AB** is raised by an increment of 5.

In Round 2, the highest temporary profit of Bidder 3 is 17 = 22 - 5. Bidder 3, following the MBR, is willing to bid on any package whose temporary profit is above 12 = 17 - 5. Bidder 3 bids on (**AB**, 5) and (**B**, 0), since the temporary profit of each package is 17 and 14, respectively. The auctioneer chooses (**AB**, 5) from Bidder 3 as the winning bid for Round 2. The process continues.

Notice, however, that Bidder 3 makes last-and-final bids on $(\mathbf{A}, 5)$ and $(\mathbf{B}, 15)$ in Round 8, because the offered prices are above the value of those packages. These bids are submitted with a discount of 5 and become $(\mathbf{A}, 0)$ and $(\mathbf{B}, 10)$, respectively. At the end of Round 9, there is no new bid. Consequently the allocation of items is finalized: Bidder 1 receives \mathbf{A} and Bidder 2 receives \mathbf{B} .

The auction proceeds to Phase II. In this phase, the auctioneer randomly chooses one of the winners as a pivotal bidder. Let us select Bidder 2 in this example. The auctioneer excludes Bidder 2's bids from the bids submitted in Round 9 and selects a new set of wining bids. The new winning bid is (AB, 20) from Bidder 3. So, the auctioneer raises the offered prices for Bidder 1 and proceeds to Round 10. In Round 10, (A,10) from Bidder 1 and (B, 10) from Bidder 3 are selected as the winning bids, and there is no new bid. Choosing Bidder 1 as the next pivotal bidder, the auctioneer goes through the same process. At the end of Round 11, since there is no new bid and no winner for the next pivotal bidder, the auction terminates.

Once the auction has ended, the auctioneer determines the price adjustment (rebate) for Bidders 1 and 2. Note that the revenue is 30 before any price adjustment. Based on the offered prices at the end of the auction, the auctioneer calculates the additional revenue that each winner generates. When Bidder 1 is pivotal, Bidder 2 receives **B** and Bidder 3 receives **A** and the revenue is 20. Thus, the additional revenue from Bidder 1 is 10 = 30 - 20, which is the price adjustment for Bidder 1. Similarly, the price adjustment for Bidder 2 is 10. All payoff information is summarized in the bottom panel of Table 1. In this example, compared to the outcomes of the Vickrey auction, *i*BEA generates higher bidder profit ($\pi_A = 10$ and $\pi_B = 20$) and less auctioneer revenue (10). This is due to the coarser grid size, $\varepsilon = 5$. In this example, if we let $\varepsilon = 1$, the outcome of *i*BEA is identical to that of Vickrey. Although *i*BEA has more desirable properties than any other ascending package auctions, it has not been tested in a laboratory setting. Past experiments on single-unit auctions show that the ascending bid auction, e.g, English clock auction, achieves higher efficiency than the sealed bid Vickrey auction, even though the solution concept is weaker (Kagel 1995). The ascending auction provides more feedbacks, which makes the optimal strategy more transparent than its sealed bid counterpart. Therefore, it is interesting to see whether this superior performance of the ascending bid auction carries over to a multi-object package bidding context.

3 Experimental Design

Our experimental design reflects both theoretical and technical considerations. Specifically, we are interested in three important questions. First, how do the Vickrey and *i*BEA auctions compare in performance? Second, how do human subjects respond to the degree of rationality in the environment? Third, do human subjects imitate bidding strategies of their robot opponents in the auction settings? We describe our experimental environment and procedures below.

3.1 The Economic Environment

In each auction, three bidders compete for four items (**A**, **B**, **C** and **D**), and thus 15 packages.⁸ In twelve out of sixteen experimental sessions, each human subject competes against two automated bidders, i.e., computer programs (or "robots"). In each session, subjects are informed that they interact with robots. Using robots gives us more control over each human subject's environment. For the human participants, this environment reduces the strategic uncertainties inherent in interactions between human bidders. Furthermore, it allows us to observe human subject reactions when they interact with robots. The latter, in itself, has important implications for e-commerce (Eisenberg 2000). In each of the remaining four sessions, five groups of three human bidders compete against each other with a random rematching protocol.

We implement a $2 \times 3 \times 2$ design for the human-robot treatments. In the first dimension, we compare the two mechanisms, the Vickrey and *i*BEA auctions. In the second dimension,

⁸These packages are A, B, C, D, AB, AC, AD, BC, BD, CD, ABC, ABD, ACD, BCD, and ABCD.

we implement three combinations of robot bidding strategies. As Sincere bidding (S) leads to efficient allocation in both mechanisms, while Random bidding (R) represents zerointelligence, we set up three different combinations of these strategies, SS, SR and RR. In SS, both robots follow Sincere bidding; in SR, one follows Sincere and the other follows Random bidding; and in RR, both follow Random bidding. As we are interested in whether a subject will imitate a certain strategy when told one of the robots follows such a strategy, we design a third dimension with two information conditions. In the low information treatment, subjects are told that they are competing against robots; however, the robot bidding strategies are not explained to the subjects. In the high information condition, we explain the robot strategies in the instructions.⁹

In addition to the human-robot treatments, we implement two human-human treatments, one for each auction mechanism, where human participants compete against each other.

We now describe bidder preferences in our experimental setting. Let V_i be the value of package *i*. The value for each item is drawn independently from a uniform distribution on $\{0, 1, 2, \dots, 10\}$. The value of a package is the sum of the values of the items in the package plus the bonus value for certain combinations of items due to synergy. In the human-robot treatments, a human bidder and the first robot derive synergy from **A** and **B**. In the experiment, we choose a CES function to represent this preference. Thus, the value of A and B together equals $V_{AB} = (V_A^{\rho} + V_B^{\rho})^{1/\rho}$. The CES function allows us to control for the degrees of complementarity and substitutability. When $\rho > 1$, $V_{AB} < V_A + V_B$. When $\rho = 1$, $V_{AB} = V_A + V_B$. When $0 < \rho < 1$, $V_{AB} > V_A + V_B$, i.e., **A** and **B** have synergy. In the experiment, we choose $\rho = 0.9$, as we are interested in the case when synergy is present.¹⁰ Similarly, the second robot derives synergy from **C** and **D**. For example, if a human bidder or Robot 1 has package **ABCD**, the value equals $V_{ABCD} = V_{AB} + V_C + V_D$. If Robot 2 has the same package, the value equals $V_{ABCD} = V_A + V_B + V_{CD}$. Since the human bidder and Robot 1 have the same preference, the environment is more competitive for each than it is for Robot 2. The same preference profile is used in the human-human treatments.

In the *i*BEA auction, we set $\varepsilon = \pounds 5$, since the smallest grid size is $\pounds 1$, and $\varepsilon = \pounds 5$ generates a reasonable speed of convergence in the lab. In Section 5, we use simulations to

⁹The instructions are quite lengthy; they are available at http://www.si.umich.edu/~yanchen/

¹⁰However, as one of our referees point out, the synergies are quite small.

compare the performance of the two mechanisms using the actual grid size of $\varepsilon = \pounds 1$ for Vickrey and $\varepsilon = \pounds 5$ for *i*BEA, as well as the same grid size of $\varepsilon = \pounds 5$ for both auctions.

3.2 Experimental Procedures

Our experiment involves 10 human subjects per session for the human-robot treatments, and 15 subjects per session for the human-human treatments. Subjects are randomly assigned to a PC terminal. At the beginning of each session, each subject is given printed instructions. After the instructions are read aloud, subjects are encouraged to ask questions. The instruction period takes, on average, 23 minutes for *i*BEA and 15 minutes for Vickrey. After receiving the instructions, subjects take a quiz designed to test their understanding of the mechanisms. At the end of the quiz, the experimenters go through the answers with the subjects. The quiz takes an average of 18 minutes for *i*BEA and 11 minutes for Vickrey. At the end of the quiz, subjects start the experiment.

In each session, each subject participates in 10 auctions, except for the two humanhuman *i*BEA sessions, where the subjects participate in three and five auctions respectively.¹¹ As we are interested in the effects of learning on bidding behavior, there are no practice auctions. At the beginning of each auction, the value for each item is randomly drawn from a uniform distribution on $\{0, 1, 2, \dots, 10\}$ for each bidder.

In the Vickrey treatments, each subject is informed of the value of the 15 packages on his screen. He can enter an integer bid for any of the 15 packages.¹² Note that a zero bid on a package is treated differently from no bid. A subject can be allocated a package on which he bids zero, but can never be allocated a package that he does not bid on. Meanwhile, each robot submits a bid on each package, following the pre-assigned strategy. A robot following the Sincere strategy bids its true value for each package, while a robot following the Random strategy randomly chooses a number between 0 and 120% of its value for each package. The upper bound is chosen based on previous experimental evidence on bidding range in single-unit Vickrey auctions (Kagel 1995). The server collects all bids from each group, computes the final allocation and payoff for each bidder and sends this information back to the bidder's screen. Each human bidder gets the following information at the end

¹¹It took 2.5 hours to finish each of the two human-human *i*BEA sessions, which was close to the upper limit we could run without causing significant fatigue among our subjects.

¹²The zTree program sets a lower bound of zero and an upper bound of 1000 for the bids.

of each auction: his allocation, his price and price adjustment, his value for the allocated package, his profit for this auction and his cumulative profit.

In the *i*BEA treatments, each subject is given a menu, which contains the following columns: the package, his value for each package, his price, his temporary profit, two check boxes (whether he wants to bid on a package, and whether he wants his bid to be the last-and-final bid) and the status of his previous bid (winning, losing, last-and-final and winning, and last-and-final and losing). The bidder may check either, both or neither of the two checkboxes. Recall that for winning bids and last-and-final bids, the checkboxes are automatically checked. During the Phase I vs. Phase II transition, the subjects see the same menu, i.e., feedback does not change in any noticeable way, and subjects are not notified when the auction moves to Phase II.

In the *i*BEA auction, a robot following the Sincere strategy adopts the MBR. A robot following the Random strategy randomly chooses a number between 0 and 120% of its value for each package, and play MBR with respect to the random valuation. Robots choose the last-and-final option for any package they place bids on that have an original negative temporary profit.

Then the auction proceeds, as described in Section 2.

[Table 2 about here.]

Table 2 presents the relevant features of the experimental sessions, including mechanisms, robot strategy profiles, information conditions, the shorthand notation for each treatment, the number of subjects for each treatment and the exchange rates.¹³ Overall, 16 independent computerized sessions were conducted in the RCGD lab at the University of Michigan from July to November 2003, and from April to June 2007. We used zTree (Fischbacher 2007) to program our experiments. Our subjects were students from the University of Michigan.¹⁴ No subject was used in more than one session, yielding a total of 180 subjects across all treatments. Each *i*BEA human-robot session lasted approximately two hours, each *i*BEA human-human session 2.5 hours, while each Vickrey session lasted approximately one hour. In addition to their auction earnings, subjects could win or lose

¹³At the end of iSS_{ℓ} and iSS_{h} , the actual earnings of subjects were low; therefore, we adjust the exchange rate to \$1 equal £1. Since these adjustments happened at the end of the experiment, and no subject was used for more than one session, we do not expect them to affect the experimental results.

¹⁴Doctoral students in Economics are excluded from participation.

money based on their quiz answers. A subject with fully correct answers gets \$5. Each mistake in the quiz costs 50 cents. The average earning (including quiz award) was \$33.1 for the Vickrey auction and \$41.6 for the *i*BEA auction. Data are available from the authors upon request.

4 Hypotheses

Based on the theoretical predictions and our experimental design, we identify the following hypotheses.

Hypothesis 1. In Vickrey auctions, bidders will bid on all packages.

Hypothesis 2. In Vickrey auctions, bidders will bid truthfully.

Hypotheses 1 and 2 are based on the dominant strategy of the Vickrey auction. The next two hypotheses are based on the theoretical predictions and our simulation results of the *i*BEA auction.

Hypothesis 3. In *iBEA* auctions, bidders will bid on all packages within the MBR-threshold¹⁵ of the maximum temporary profit and only bid on these packages.

Hypothesis 4. In *iBEA* auctions, bidders will choose the last-and-final option for all packages with negative temporary profits.

We now formulate hypotheses which compare the performance of the two mechanisms. As *i*BEA is an ascending version of Vickrey, we expect, *ex ante*, that the two mechanisms will generate the same bidder profit, auctioneer revenue and efficiency.

Hypothesis 5. Vickrey and iBEA will generate the same amount of bidder profit.

Hypothesis 6. Vickrey and iBEA will generate the same amount of auctioneer revenue.

Hypothesis 7. Vickrey and iBEA will generate the same efficiency.

Furthermore, as the mechanisms are quite complex, we expect the human bidder to learn from robot strategies in the high information treatments.

¹⁵The MBR-threshold is defined and computed based on simulation results in Section 5.

5 Results

In this section, we first examine individual bidder behavior in Vickrey and iBEA auctions. We then compare the aggregate performance of the two auctions.

5.1 Individual Behavior in Vickrey Auctions

Our Vickrey auction experiment consists of 90 subjects. Each subject independently plays 10 auctions with two robots in one of the six human-robot treatments. In the human-human treatment, in each session, 15 subjects are randomly rematched into groups of three and compete within their group in each of the ten auctions. In each auction, a subject can bid on any of the 15 packages at any price between £0 and £1000.

Unlike the single item case, the strategy in a multi-item Vickrey auction with package bidding has two dimensions. The first dimension is whether to bid on a package. The second is how much to bid on a package if one decides to bid on it. A bidder's strategy on either dimension affects his profit, as illustrated in Figure 1.

[Figure 1 about here.]

Figure 1 presents simulated profits for the human bidder under the three different human-robot environments in the Vickrey auction. The top panel (a) presents results for the environment with two sincere robots. The middle panel (b) presents results for the environment with one sincere and one random robot. The bottom panel (c) presents results for the environment with two random robots. The horizontal plane consists of the two strategy dimensions, the probability of bidding on a package, and the bid/value ratio for a package. In the simulation, we generate 20,000 hypothetical auctions, each of which consists of independent draws of the values for items A, B, C and D from the uniform distribution on $\{0, 1, 2, \dots, 10\}$. In our hypothetical auctions, the preferences for the human and robot bidders, as well as the auction rules, are identical to the experimental environment. For each combination of the Probability of Bidding (drawn from the interval [0, 1] with a grid size of 0.05) and Bid/Value Ratio (drawn from the interval [0, 2] with a grid size of 0.1), we compute the average profit for the human bidder across these 20,000 auctions, and report it on the vertical axis. Comparing the profit from each strategy combination across environments, we find that, Sincere-Sincere is the most competitive environment, with the lowest profit level for the human bidder occurring at any given combination of the probability of bidding and the bid/value ratio. The Sincere-Random environment is the next most competitive, followed by the Random-Random environment. In each environment, the highest profit is achieved when the probability of bidding is 1, and the bid/value ratio is also 1, i.e., bidding on every package and bidding one's true value for each package, which is the dominant strategy.

[Table 3 about here.]

Table 3 presents results for the human bidders' bidding decisions, averaged across all treatments. The Active Bids column presents the ratio of positive and zero bids over 900, which is the total number of bids if every bidder bids on every package. Note a zero bid is an active bid, while no bid is inactive. The Bid/Value Ratio is the mean ratio of the bid to the value of a package. The next three columns present the proportion of Truthful Bidding, Overbidding and Underbidding among active bids, respectively.

Even though bidding on every package is a weakly dominant strategy, the results in the second column of Table 3 show that the proportion of bids never reaches 100% on any package. Furthermore, participants bid on packages containing items AB more frequently than they do on other packages. We now investigate which factors induce a higher proportion of active bids.

[Table 4 about here.]

Table 4 presents results from the four probit models with robust clustering at the individual level. The dependent variable is Active Bids, a dummy variable, which equals one if a bidder places an active bid on a package and zero otherwise. While specifications (1) and (2) examine factors affecting the likelihood of active bids in the human-robot treatments, specifications (3) and (4) include the two sessions of human-human data. The independent variables include Value of a package, a dummy variable D_{synergy} , which equals one if a package contains both of the items that have the synergy,¹⁶ and zero otherwise, and a dummy variable D_{HS} , which equals one if the information condition is High and there is at least one sincere robot bidder, and zero otherwise. The reason for including the latter is that

¹⁶In the human-human treatments, for Bidder 3, \mathbf{C} and \mathbf{D} have the synergy. For Bidders 1 and 2, these items are \mathbf{A} and \mathbf{B} .

participants might learn from the sincere robot strategy. Even though we are explicit in the instructions for all High Information treatments that robots bid on all packages, participants might ignore both parts of the random robot strategy, as the second part is obviously not optimal. Indeed, replacing the last dummy with a dummy D_H , which equals one with High Information and zero otherwise, results in an insignificant coefficient. Lastly, the dummy variable D_{human} equals one for the human-human treatment, and zero otherwise.

Result 1 (Whether to Bid on a Package). Bidders are significantly more likely to bid on packages with higher values. In addition, they are more likely to bid on packages with synergistic items. Within the human-robot treatments, the proportion of active bids increases significantly in treatments with at least one sincere robots and high information.

Support. In Table 4, the coefficients are probability derivatives. In specification (4), increasing Value by 1 increases the likelihood of bidding on this package by 0.7%. If a package contains the synergistic bundle, the likelihood that a subject bids on this package is increased by 7.7%. In specification (1), compared to other information conditions, subjects increase the likelihood of bidding on packages by 16.4% with at least one sincere robot and high information. All coefficients are significant at the one- or five-percent level.

Result 1 indicates that bidders are significantly more likely to bid on high value packages and those with synergistic bundles. Furthermore, bidders imitate the Sincere Robot, but not the Random Robot. The coefficient for the dummy variable for the human-human treatment, D_{human} , is positive and weakly significant (at the 10% level), indicating that bidders are 10.7% more likely to place an active bid when competing against other humans than when competing against robots.

We now explore the second dimension of bidding strategy, how much to bid on a package. To investigate how much participants bid on a package in a Vickrey auction, we use a structural approach based on Hypothesis 2, which proposes that bidding one's true valuation is a weakly dominant strategy. To test this hypothesis, we use an OLS regression with clustering at the individual level. In the first specification, we use Bid as the dependent variable, and Value as the only independent variable. We do not include a constant because of the theoretical prediction. In the second specification, we add Auction, Cumulative Profit, $D_{\rm synergy}$ and $D_{\rm HS}$ as independent variables. The variable Auction denotes the number of auctions in a session. Thus, this variable captures any learning effect. For each specification,

we run two-sided Wald tests of the null hypothesis of bids being equal to values against the alternative hypothesis of bids not being equal to values. The results are presented in Table 5.

[Table 5 about here.]

The results in Table 5 indicate that, on average, bidders tend to underbid, rather than overbid, in Vickrey auctions. In specification (1), the coefficient for Value is 0.919, significantly less than one (p < 0.01, one-sided Wald test). In a similar regression conducted by Isaac and James (2000), the coefficient for Value is 0.95 and not statistically different from one.¹⁷ To classify bidders, we repeat the first specification in Table 5 for each bidder. We then perform the Wald test for the null hypothesis that the coefficient on Value is 1 and subsequently classify bidders into the following groups.

- 1. Underbidder: If we can reject the hypothesis of truthful bidding at the 5% level and the coefficient is below 1.
- 2. Truthful Bidder: If we cannot reject the hypothesis of truthful bidding at the 5% level.
- Overbidder: If we can reject the hypothesis at the 5% level and the coefficient is above
 1.

We now summarize the analysis of bidding behavior in Vickrey in the following result.

Result 2 (Bid Price in Vickrey). Bidders in a Vickrey auction, on average, bid 91.9% of their true value. The coefficient of Value is significantly less than unity. Of our participants, 73% can be classified as underbidders, 6.7% as truthful bidders and 20.0% as overbidders.

Support. Table 5 presents the OLS regression results for our Vickrey auctions. The coefficient estimates show the amount subjects bid compared to their valuations. Robust standard errors in parentheses are clustered at the individual level. A one-sided Wald test of the null hypothesis of bid being equal to value yields p-value less than 0.01. The classification of bidders comes from regressions at the individual level. The average R^2 of individual regressions is 0.939, with a standard deviation of 0.125.

 $^{^{17}}$ Isaac and James (2000) also include a constant, which is estimated to be -0.19 and not statistically different from zero.

By Result 2, we reject Hypothesis 2. Most previous laboratory studies of single-unit Vickrey auctions find that bidders tend to overbid in such environments (Kagel 1995). In multi-unit uniform price auctions, bidders tend to overbid on the first unit and underbid on the second unit, which is consistent with the theoretical prediction of demand reduction (Kagel and Levin 2001). Our finding that most bidders either underbid or bid their true value in Vickrey auctions is in stark contrast with previous experimental results. Our study presents empirical evidence that the "robust" finding of overbidding in single-unit Vickrey auctions does not carry over to package Vickrey auctions.

In addition to studying overall bidding behavior, we examine how learning may impact bidding behavior. To do so, we first define *bidding pattern* as a combination of two different measures, the Number of Active Bids¹⁸ in an auction and the Bid/Value Ratio. To identify the effects of prior experience while minimizing individual-specific characteristics on bidding behavior, we use the difference-in-difference analysis. First, for each auction, we take the difference of the Number of Active Bids between the current and the previous auctions. We then classify all observations into two groups, a winner group where subject(s) won a package in the previous auction, and a loser group, where they did not. Finally, we compare the difference between the two groups. We analyze the Bid/Value Ratio in a similar way.

Result 3 (Effect of Prior Experience on Bidding). Losers in a previous auction are significantly more likely to change their number of active bids, compared to winners. Furthermore, losers increase their bid/value ratio, while winners decrease their bid/value ratio. The difference is significant at the one-percent level.

Support. A loser in a previous auction changes his number of active bids by 1.95, on average (with a standard error of 0.176), while a winner in the previous auction changes his number of active bids by only 1.19 (with a standard error of 0.097). The difference is statistically significant at the level of 1%. The robust standard error clustered by subject is 0.242 that yields a p-value of 0.002. Furthermore, a loser in a previous auction increases his bid/value ratio by 0.129 on average (with a standard error of 0.034), while a winner in the previous auction decreases his bid/value ratio by 0.064 (with a standard error of 0.093). The difference is statistically significant at the 1% level. The robust standard error clustered at the individual level is 0.052 which yields a p-value less than 0.001.

¹⁸This is equivalent to the likelihood of bidding when the total number of packages is fixed.

Result 3 indicates that participants in our study learn from prior experience. The directions of change for winners and losers are intuitive, yet indicate that the dominant strategy is not transparent to a substantial number of participants, who adjust their behavior by trial and error.

To understand the individual learning dynamics in our Vickrey auctions, we also investigate individual learning through two learning models, the reinforcement learning model (Erev and Roth 1998) and the payoff assessment learning model (Sarin and Vahid 1999). Both models have been shown to track human learning behavior fairly well in a variety of games, such as relatively simple games (Erev and Roth 1998), games with complete information (Chen and Gazzale 2004) and of limited information (Chen and Khoroshilov 2003). However, we find that neither tracks learning dynamics well in the more complex Vickrey auctions.

Overall, the most surprising finding in our Vickrey auctions is that most bidders either underbid or bid their true value. Even though the dominant strategy is not transparent, our subjects tend to adjust their behavior by trial and error. These findings in individual behavior will translate into interesting aggregate performance measures, which we analyze in Section 5.3.

5.2 Individual Behavior in *i*BEA

Our *i*BEA experiment consists of 90 subjects. In the human-robot treatments, each subject independently plays 10 auctions with two robots. In the human-human treatments, in each session, 15 subjects are randomly rematched into groups of three at the beginning of each auction. As the *i*BEA auctions with a random rematching protocol take much longer to complete, the number of auctions is three and five, respectively.

To analyze our *i*BEA auction results, we define *adjusted temporary profit* as the temporary profit of a package after a subject considers whether to check the last-and-final option. For packages with the last-and-final option checked, the temporary profit is increased by £5. The myopic best response (MBR) strategy in an *i*BEA auction states that bidders should bid on packages within ε of the maximum adjusted temporary profit. Among packages a participant can actively bid on,¹⁹ we group packages by their adjusted temporary profits in

¹⁹Recall that last-and-final packages and winning packages from previous rounds are automatically checked in the current round; therefore, a participant cannot act on them. Consequently, we do not consider them

each round of the auction.

At each round, a package is called a *MBR package* if its adjusted temporary profit is within the MBR-threshold of the maximum adjusted temporary profit, where the MBRthreshold is computed such that bidding on all packages within this threshold maximizes the bidder's expected profit.

Recall that Sincere robot's threshold is set to $\pounds 5$. Given the robot strategies, the threshold for human bidders are computed through simulation presented in Figure 2.

[Figure 2 about here.]

Figure 2 presents simulated profits for the human bidder in the three human-robot treatments (iSS, iSR and iRR) in the *i*BEA auction. The horizontal plane consists of the two strategy dimensions. The first, MBR threshold, determines which packages are *MBR packages* for the human bidder, and the second is the probability of bidding on those MBR packages. For each combination of the MBR threshold (drawn from the interval [1, 15] with a grid size of 1) and the Probability of Bidding (drawn from the interval [0, 1] with a grid size of 0.05), we generate 20,000 hypothetical auctions, each of which consists of independent draws of the values. Then we compute the average profit for the human bidder across those 20,000 auctions and report it on the vertical axis. As indicated in the figure, the MBR threshold is 6 for the iSS and iRR treatments, and 7 for the iSR treatment.

[Figure 3 about here.]

Figure 3 presents the proportion of human bidders that bid on MBR and non-MBR packages among all packages a participant can actively bid on, in each of the six different treatments. Comparing the low and high information treatments within SS, SR and RR, the proportion of MBR bids to non-MBR bids substantially increases in SS. This increase is consistent with the hypothesis that humans learn from robot strategies. We now use probit models to formally check our impression from Figure 3.

[Table 6 about here.]

Table 6 presents seven probit specifications which examine factors affecting the likelihood of bidding on a package. In all specifications, the dependent variable is PlaceBid, a dummy

as part of the choice set for the subject.

variable which equals one if a bid is placed on a package and zero otherwise. The independent variables are Temporary Profit, D_{synergy} , D_{H} and MBR in specifications (1), (3) and (5). In specifications (2), (4) and (6), we add an independent variable, $D_{\text{H}} \times \text{MBR}$, which equals one if the information condition is high and the package belongs to the set of MBR packages, and zero otherwise. This variable controls for the information condition on MBR package bidding. In specification (7), we only include two independent variables, Temporary Profit and D_{synergy} . We summarize the analysis below.

Result 4 (Bidding Decision in *i*BEA). In our *i*BEA auctions, we find that bidders are significantly more likely to bid on packages with a higher temporary profit. We also find that additional information on robot strategies induces significantly more bids on MBR packages under SS.

Support. In Table 6, the coefficients on Temporary Profit are positive and significant in all seven specifications. The coefficients on $D_{\rm H}$ are not significant in any specifications. The coefficient of $D_{\rm H} \times {\rm MBR}$ is positive and significant in (2).

Result 4 confirms our intuition from Figure 3 that knowledge of robot strategies significantly impacts human bidding behavior. In particular, the Sincere Robot induces more MBR bids. This suggests that, in a complex auction, participants might not be able to figure out the optimal bidding strategy, and thus are susceptible to learning. Therefore, in the actual implementation of an *i*BEA auction, teaching bidding strategies may make bidding more effective.

We now explore specifically whether subjects learn the robot strategy regarding the last-and-final option. In all *i*BEA human-robot treatments, all robots follow the equilibrium strategy in checking the last-and-final option, i.e., they check this option for a package whose temporary profit is negative. In our high information treatments, subjects are explicitly told when robots check the last-and-final option. We are interested in two questions. First, do subjects learn to use this option? Second, when they use this option, do they use it at the right time?

[Table 7 about here.]

Table 7 presents summary statistics on the mean temporary profit of all last-and-final bids in a treatment, the number of such bids where the temporary profits are negative, or non-negative, the total number of last-and-final bids and the proportion of negative lastand-final bids. The average temporary profit is £2.43 for all treatments, £2.13 for high information treatments and £2.73 for low information treatments. The difference between the high and low information treatments is not significant. This indicates that subjects tend to check this option earlier than would be optimal.

[Table 8 about here.]

Table 8 presents OLS regression results. The dependent variable is Temporary Profit of Last-and-Final Bids, while the independent variables include Round, and a dummy variable for high information, D_H . The coefficients of Round is negative and significant in all three specifications, indicating that participants learn to check the last-and-final option with less temporary profits over time.

Overall, in our *i*BEA auctions, consistent with the MBR strategy, bidders are more likely to bid on packages with higher temporary profits, however, they check the last-andfinal option earlier than would be optimal. Furthermore, bidders seem to imitate bidding strategies from the Sincere robots, and not the Random robots.

5.3 Aggregate Performance of the Two Mechanisms

We now examine the aggregate performance of the two mechanisms in three aspects: bidder profit, auctioneer revenue and efficiency.

In examining bidder profit, we look at both human bidder profit and aggregate human and robot profit in each treatment. Auctioneer revenue follows the standard definition.

In single-unit auctions, efficiency is often measured by a ratio of the number of auctions where the object goes to the bidder with the highest valuation to the total number of auctions. In the multi-unit context, this definition is not applicable. Instead, we use the definition of efficiency developed by Kagel and Levin (2001). In this definition, efficiency of an auction is the ratio of the total surplus of the allocation to the highest possible surplus among all possible $3^4 = 81$ allocations, where total surplus is the sum of bidder profit and auctioneer revenue. Then, for each auction, we normalize the ratio by the average surplus of 81 allocations as follows,

$$Efficiency = \frac{\text{the actual total surplus} - \text{average surplus}}{\text{the highest possible surplus} - \text{average surplus}}$$
(3)

When we compare performance of the two mechanisms, we need to adjust the price increments such that they are comparable. Recall that, in the experiment, the price increment in the *i*BEA auction is £5 due to the length of the auction, while in the Vickrey auction we use a grid size of £1. To address the effect of grid size on performance, for each performance measure, we present two versions for the Vickrey auction, one representing the observed results and a second representing the proxy results. In computing the latter, we input the actual bids submitted under the Vickrey mechanism, and use a proxy agent for each bidder. The proxy agent uses the actual bids as values, and follows the MBR strategy. Allocations are determined using the *i*BEA algorithm with a grid size of £5. When we use the proxy agents for the Vickrey auction (hereafter Vickrey proxy), *i*BEA and Vickrey auctions are more comparable in aggregate performance measures.

[Table 9 about here.]

Table 9 presents summary statistics on the average human profit, total bidder profit, auctioneer revenue and efficiency under each auction mechanism. This comparison shows that, while the *i*BEA auction yields higher total bidder profit and the Vickrey auction achieves higher revenue, the efficiency comparison is not clear-cut. Since session average does not control for a number of exogenous factors, we use the following specifications to model factors affecting the performance of each mechanism.

[Table 10 about here.]

Table 10 presents four OLS specifications. The dependent variable in each specification is (1) Human Profit, (2) Total Profit, (3) Revenue and (4) Efficiency. The independent variables are Vickrey Proxy, which equals 1 for the Vickrey auction with a coarser grid size and 0 for *i*BEA, $D_{\rm H}$, the number of random robots, quiz score and a constant. The top panel includes all human-robot treatments, while the bottom panel only include the SS treatments. Note that we use the Vickrey proxy in computing each of these measures, so that it is comparable to the *i*BEA auction. We summarize our findings below.

Result 5 (Bidder Profit). Total bidder profit is significantly higher in the *i*BEA auction than in the Vickrey auction. Furthermore, the number of random robots significantly increases human profit, and weakly decreases total profit.

Support. In Table 10, for all human-robot treatments (top panel), for specification (1), the coefficient for the number of random robots is positive and significant. In specification (2), the coefficient for Vickrey Proxy is negative and significant, while the coefficient for the number of random robots is negative and weakly significant at the 10% level. For the SS treatments (bottom panel), in specification (2), the coefficient for Vickrey Proxy is negative and significant.

By Result 5, we reject Hypothesis 5. As an ascending version of the Vickrey auction, the *i*BEA auction generates significantly higher bidder profit. Consistent with our simulations, the number of random robots significantly increases human profit.

Result 6 (Revenue). Vickrey auctions generate significantly higher revenue than *i*BEA auctions. In addition, the number of random robots significantly decreases revenue in both auctions. For the SS treatments, a higher quiz score significantly increases revenue.

Support. In Table 10, for specification (3) (top and bottom panels), the coefficient for Vickrey Proxy is positive and significant. On the top panel, the coefficient for the number of random robots is negative and significant, while on the bottom panel, the coefficient for quiz score is positive and significant.

By Result 6, we reject Hypothesis 6. The fact that more than 60% of the last-and-final bids in *i*BEA have positive temporary profits (Table 7) indicates that human bidders did not bid optimally in the *i*BEA auction, which partially explains Result 6.

[Figure 4 about here.]

We next investigate the efficiency comparisons between the two auctions. Figure 4 contains six panels. The top two panels present the distribution of actual observed efficiencies in all auctions, pooling across all treatments. The left panel is the efficiency distribution under the Vickrey auctions, and the right panel is the efficiency distribution under the *i*BEA auctions. The middle two panels present the same information comparing *i*BEA and Vickrey proxy, while the bottom two panels compares *i*BEA and Vickrey proxy in the SS environment only.

Result 7 (Efficiency). Vickrey auction generates the same efficiency as the *i*BEA auction. However, 26.3% of the *i*BEA auctions and 16.5% of the Vickrey auctions achieve 100% efficiency. The difference is significant at the 10% level. Furthermore, the number of random robots significantly decreases efficiency.

Support. In Table 10, for specification (4), the coefficient for Vickrey Proxy is insignificant for all human-robot treatments (top panel) and SS treatments (bottom panel). The coefficient for the number of random robots is negative and significant. In the human-robot treatments, we compute the proportion of auctions achieving 100% efficiency for each subject, which generates one independent observation for each subject, and compare the performance of the two mechanisms. A one-sided proportion of t-test yields p = 0.095, rejecting the null hypothesis of equal proportion in favor of the alternative hypothesis that *i*BEA generates a higher proportion of auctions achieving 100% efficiency.

By Result 7, we fail to reject Hypothesis 7. Our results show that, with package bidding, the Vickrey auction achieves the same efficiency as does the ascending *i*BEA auction. However, the latter generates a higher proportion of auctions which achieve 100% efficiency.

While Results 5 to 7 compare the aggregate performance of the two auctions for the human-robot sessions, we now investigate any systematic difference between the human-robot and human-human treatments.

[Table 11 about here.]

Table 11 reports the p-values for Wilcoxon rank sum tests comparing the human-human with human-robot treatments for the iBEA (upper panel) and Vickrey (lower panel) auctions respectively. For each comparison, while the null hypothesis is that the performance of the human-human treatment is the same as that of the human-robot treatment, the alternative hypothese are stated in row 2 in brackets. Recall that the session average performance measures are reported in Table 9. For the rank sum tests, we treat each session (bidder) as an independent observation for the human-human (human-robot) treatments. Results are summarized below.

Result 8 (Human-human vs. human-robot). In both the Vickrey and *i*BEA auctions, the human-human treatments generate significantly higher human profit and significantly lower revenue compared to the corresponding human-robot treatments. Efficiency comparisons are not significantly different except in the SS environment, where efficiency is significantly higher in the human-robot treatments.

Support. In Table 11, one-sided Wilcoxon ranksum tests yield p < 0.05 for human profit (except in vRR_l) and total profit (except in iSS_h), and p < 0.05 for revenue (except in vRR_h), rejecting H_0 in favor of H_1 . While two-sided Wilcoxon ranksum tests yields p > 0.05 for efficiency comparisons in all other human-robot treatments, in the four SS treatments, one-sided Wilcoxon ranksum tests yield p < 0.05 rejecting H_0 in favor of H_1 that the human-robot treatments generate significantly higher efficiency.

Result 8 indicates that, in the human-human treatments in both auctions, bidders behave as if they collude to exploit auctioneer revenue. Furthermore, the human-robot treatment in the SS environment generates significantly higher efficiency than the corresponding human-human treatment, indicating two sources of inefficiency in our experiment, i.e., introducing random robots or other human bidders. However, the way in which a second and third human bidders cause inefficiency is different from that in which random robots do so. More specifically, the former results in much greater profit to the human bidders and much lesser revenue to the auctioneer.

6 Conclusion

Package auctions have become increasingly popular in procurement and complex resource allocation contexts and thus have stimulated a large body of theoretical research in combinatorial auctions in economics and computer science. Because of the concerns over the deficiencies of Vickrey auctions, several new ascending package auctions have been proposed to implement efficient allocations under various assumptions. One of the most prominent new ascending package auctions is the *i*BEA auction, which achieves approximately efficient allocation and implements Vickrey payments under minimal assumptions on preferences.

As a first step in using this auction as an actual economic process that solves naturally occurring problems, we observe the performance of the *i*BEA auction in the context of simple situations that can be created in a laboratory. We then assess its performance relative to a natural and important benchmark, the sealed-bid Vickrey auction.

As the first experimental study of the *i*BEA auction in comparison with the Vickrey auction, we use a simple environment where each human bidder competes against two robots with different levels of bidding intelligence. This implementation creates an environment free from the strategic uncertainties inherent in interactions between human bidders. As a

robustness check, we also implement both mechanisms among all human bidders.

Our experiment yields several surprising findings. First, unlike the single-unit Vickrey auction where bidders tend to overbid in the laboratory, most of our bidders either underbid or bid their true value. A simple dynamic adjustment model captures the learning dynamics reasonably well. Second, in terms of aggregate performance, while bidder profit is significantly higher in the *i*BEA auction, the Vickrey auction generates significantly higher revenue than does the *i*BEA auction. While the two auctions achieve the same efficiency, a higher proportion of *i*BEA auctions achieves 100% efficiency than does the Vickrey auction. Lastly, we find that when human bidders compete against robots in a complex environment, they learn from their robot opponents when the robot strategies are intelligent (e.g., myopic best responses).

With this first laboratory study of the *i*BEA auction, we identify several issues that warrant further empirical study. The first issue is the tradeoff between the speed and efficiency of the auction as manipulated through setting different price increments. The larger the auctioneer sets the price increment, the faster the auction converges. However, the final allocation might be further away from the efficient allocation. In an auction with a large number of bidders, we expect the second phase of the *i*BEA auction to take longer. Thus, it is important to quantify the speed-efficiency tradeoff. The second issue is whether bidders can detect the second phase of the auction and thus collude to increase bidder profit. Our human-human sessions result in significantly higher bidder profit compared to the corresponding human-robot sessions, indicating that collusion might be a concern even in our simple environment. Lastly, alternative implementations of the *i*BEA auction might improve its performance.²⁰ For example, the experimenter can teach or recommend the MBR strategies to the bidders, or she can show where more profit is available in addition to report the temporary profit in bundles reported in a round. Both methods can help the bidder better optimize given the time limits in an experiment or actual implementation.

Empirical investigations of package auctions have important implications not only in procurement and privatization, but also potentially in the allocation of scarce equipment time in large scientific collaboratories. Past studies, such as Kwasnica et al. (2005), show that, with synergies, package auctions tend to outperform auctions without package bidding. The natural next step is to evaluate the pool of package auction mechanisms under a variety

 $^{^{20}}$ We thank an anonymous referee for these suggestions.

of environments in the laboratory, select those which perform robustly well, and test them in a field setting.

In selecting the pool of package auctions for practical implementation, iterative auctions are important for a number of reasons. First, they are potentially more transparent and should provide better performance in practice. Second, in the real world implementations, when valuations are too complex to be enumerated with a sealed bid Vickrey auction, iterative auctions may perform much more effectively because price discovery guides bidders in recognizing which parts of their demand to report. When the value of an asset requires costly investments to be discovered, ascending price auctions generate higher revenue than their static counterparts (Compte and Jehiel 2007). Similarly, Parkes (2005) shows that ascending-price auctions can achieve better efficiency than sealed-bid auctions when incorporating costly preference elicitation. These theoretical results point to fruitful areas of future research in evaluating package auction mechanisms. For example, in the laboratory, we can require participants to pay for every additional bundle that they want to bid for, where payments are used as proxies for cognitive load. Cognitive load is an important factor in choosing mechanisms for complex resource allocation, but to our best knowledge it has not been incorporated in any of the experimental auction studies.

		bidder 1		k	bidder 2		bidder 3			
Pac	kage	А	В	AB	А	В	AB	А	В	AB
Va	lue	10	0	10	0	30	30	4	14	22
Phase	Round	(B	id, Pr	ice)	(B	id, Pı	rice)		(Bid, Pric	e)
Ι	1	(A, 0)		(AB, 0)	(B, 0)		(AB, 0)			(AB, 0)
Ι	2	(A, 0)		(AB, 0)	(B, 0)		(AB, 0)		(B, 0)	(AB, 5)
Ι	3	(A, 5)		(AB, 5)	(B, 5)		(AB, 5)		(B, 0)	(AB, 5)
Ι	4	(A, 5)		(AB, 5)	(B, 5)		(AB, 5)		(B, 5)	(AB, 10)
Ι	5	(A, 5)		(AB, 5)	(B, 5)		(AB, 5)	(A, 0)	(B,10)	(AB, 15)
Ι	6	(A, 5)		(AB, 5)	(B, 10)		(AB, 10)	(A, 0)	(B,10)	(AB, 15)
Ι	7	(A, 5)		(AB, 5)	(B, 15)		(AB, 15)	(A, 0)	(B, 10)	(AB, 15)
Ι	8	(A, 5)		(AB, 5)	(B, 15)		(AB, 15)	(A, 0)	(B, 10)	(AB, 20)
Ι	9	(A, 5)		(AB, 5)	(B, 15)		(AB, 15)	(A, 0)	(B, 10)	(AB, 20)
II		(A, 5)		(AB, 5)	(B,15)		(AB,15)	(A, 0)	(B, 10)	(AB, 20)
II	10	(A,10)		(AB, 10)	(B,15)		(AB,15)	(A, 0)	(B, 10)	(AB, 20)
II		(A,10)		(AB,10)	(B, 15)		(AB, 15)	(A, 0)	(B,10)	(AB,20)
II	11	(A,10)		(AB,10)	(B,20)		(AB,20)	(A, 0)	(B,10)	(AB,20)

1. Boldface indicates the winning bid after the current round.

2. Italics indicate bids with the last-and-final option.

3. (Bid, Price) indicates that the (Bid, Price) pair is excluded.

iBEA	Package	Price	Piv. Revenue	Rebate	Final Price	Profit
bidder 1	А	10	20	10	0	10
bidder 2	В	20	20	10	10	20
bidder 3	-	-	-	-		0
auctioneer	-	-	-	-		10
Vickrey	Package	Price	Piv. Revenue	Rebate	Final Price	Profit
Vickrey bidder 1	Package A	Price 10	Piv. Revenue 34	Rebate 6	Final Price	Profit 6
Vickrey bidder 1 bidder 2	Package A B	Price 10 30	Piv. Revenue 34 24	Rebate 6 16	Final Price 4 14	Profit 6 16
Vickrey bidder 1 bidder 2 bidder 3	Package A B -	Price 10 30 -	Piv. Revenue 34 24 -	Rebate 6 16 -	Final Price 4 14	Profit 6 16 0

Note: Piv. Revenue refers to total revenue when bidder i is excluded.

Table 1: A Simple Example of an *i*BEA Auction Process (Grid Size $\varepsilon = 5$)

Mechanism	Robot Strategy	Information	Notation	# of Subjects	Exchange Rate
		High	iSS_h	10	2
iBEA	Sincere, Sincere	Low	iSS_ℓ	10	2
		High	iSR_h	10	1
iBEA	Sincere, Random	Low	iSR_ℓ	10	1
		High	iRR_h	10	1
iBEA	Random, Random	Low	iRR_ℓ	10	1
		n/a	iHumans	15	1
iBEA	(all humans)	n/a	iHumans	15	1
		High	vSS_h	10	1.25
Vickrey	Sincere, Sincere	Low	vSS_ℓ	10	1.25
		High	vSR_h	10	1.25
Vickrey	Sincere, Random	Low	vSR_ℓ	10	1.25
		High	vRR_h	10	1.25
Vickrey	Random, Random	Low	vRR_ℓ	10	1.25
		n/a	vHumans	15	1.25
Vickrey	(all humans)	n/a	vHumans	15	1.25

 Table 2: Features of Experimental Sessions

Package	Active Bids	Bid/Value	Underbidding	Truthful Bidding	Overbidding
Α	0.722	0.925	0.606	0.237	0.157
в	0.682	0.906	0.586	0.259	0.155
\mathbf{C}	0.677	1.045	0.612	0.243	0.144
D	0.682	0.834	0.599	0.256	0.145
\mathbf{AB}	0.859	0.961	0.648	0.176	0.176
\mathbf{AC}	0.738	0.912	0.682	0.178	0.140
AD	0.733	0.872	0.695	0.165	0.139
\mathbf{BC}	0.730	0.902	0.682	0.166	0.152
BD	0.730	0.891	0.688	0.161	0.151
$\mathbf{C}\mathbf{D}$	0.766	0.881	0.668	0.196	0.136
ABC	0.840	0.935	0.671	0.161	0.168
ABD	0.814	0.918	0.685	0.154	0.161
ACD	0.777	0.913	0.688	0.172	0.140
BCD	0.758	0.911	0.674	0.177	0.148
ABCD	0.831	0.952	0.670	0.162	0.168

Table 3: Proportion of Active Bids, Bid/Value Ratio and Proportion of Under-, Truthful, and Overbidding in the Vickrey Auction

Dependent Variable: Active Bids						
	Human-Rob	oot Treatments	All Vickrey Treatments			
	(1)	(2)	(3)	(4)		
Value	0.008	0.008	0.007	0.007		
	$(0.002)^{***}$	$(0.003)^{***}$	$(0.002)^{***}$	$(0.002)^{***}$		
D _{synergy}	0.085	0.086	0.076	0.077		
	$(0.021)^{***}$	$(0.021)^{***}$	$(0.017)^{***}$	$(0.018)^{***}$		
D_{HS}	0.164		0.149			
	$(0.072)^{**}$		$(0.061)^{**}$			
D _H		-0.017		-0.016		
		(0.077)		(0.074)		
$\mathrm{D}_{\mathrm{human}}$			0.107	0.055		
			$(0.061)^*$	(0.071)		
Observations	9000	9000	13500	13500		

1. Coefficients are probability derivatives.

2. Robust standard errors in parentheses are adjusted for clustering at the individual level.

3. Significant at: ** 5% level; *** 1% level.

Table 4: Probit: Factors Affecting the Likelihood of Active Bids in the Vickrey Auction

	Dependent	Variable: Bid
	(1)	(2)
Value	0.919	0.906
	$(0.024)^{***}$	$(0.022)^{***}$
Auction		0.137
		(0.099)
Cum. Profit		-0.055
		$(0.026)^{**}$
D _{synergy}		0.546
		$(0.223)^{**}$
D_{HS}		-0.254
		(0.550)
Constant		0.111
		(0.617)
Observations	10205	10205
R-squared	0.81	0.51

1. Robust standard errors in parentheses are adjusted for clustering at the individual level.

2. Significant at: * 10% level; ** 5% level; **
* 1% level.

Table 5: OLS: Bidding Decision in the Vickrey Auction

	Dependent	Variable: Pla	ceBid				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	iSS	iSS	iSR	iSR	iRR	iRR	iHumans
Temp. Profit	0.008	0.008	0.008	0.008	0.019	0.019	0.017
	$(0.002)^{***}$	$(0.002)^{***}$	$(0.003)^{***}$	$(0.003)^{***}$	$(0.004)^{***}$	$(0.004)^{***}$	$(0.003)^{***}$
$\mathbf{D}_{\mathrm{synergy}}$	0.029	0.027	0.047	0.046	0.066	0.066	0.029
	$(0.014)^{**}$	$(0.014)^{**}$	$(0.015)^{***}$	$(0.015)^{***}$	$(0.021)^{***}$	$(0.021)^{***}$	$(0.017)^{*}$
MBR	0.12	0.038	0.103	0.14	0.075	0.053	
	$(0.039)^{***}$	$(0.020)^{*}$	$(0.043)^{**}$	$(0.064)^{**}$	$(0.018)^{***}$	$(0.026)^{**}$	
D _H	0.065	-0.016	-0.11	-0.062	-0.05	-0.073	
	(0.056)	(0.051)	(0.073)	(0.077)	(0.085)	(0.083)	
$\mathrm{D_{H} \times MBR}$		0.174		-0.08		0.046	
		$(0.065)^{***}$		(0.067)		(0.041)	
Observations	23841	23841	23454	23454	21051	21051	14497

1. Coefficients are probability derivatives.

2. Robust standard errors in parentheses are adjusted for clustering at the individual level.

3. Significant at: * 10% level; ** 5% level; **
* 1% level.

Table 6: Probit: Likelihood of Bidding in the *i*BEA Auction

Treatment	Mean Temp. Profit	Negative	Non-Negative	Total	Negative/Total
iSS_ℓ	4.08	38	81	119	31.9%
iSR_ℓ	1.93	100	113	213	46.9%
iRR_ℓ	2.44	45	127	172	26.2%
iSS_h	1.36	128	108	236	54.2%
iSR_h	3.11	46	104	150	30.7%
iRR_h	2.29	92	138	230	40.0%
iHumans	3.22	28	91	119	23.5%
Pooled	2.43	477	762	1239	38.5%

Table 7: Last-and-Final Summary Statistics for the *i*BEA Auction

Dependent Variable: Temp. Profit of Last-and-Final Bids						
	SS	SR	RR			
Round	-0.73	-0.67	-0.57			
	$(0.141)^{***}$	$(0.165)^{***}$	$(0.170)^{***}$			
D_{H}	-3.10	1.32	0.28			
_	(2.031)	(1.819)	(1.826)			
Constant	9.15	6.19	5.86			
	$(2.566)^{***}$	$(2.117)^{***}$	$(1.966)^{***}$			
Observations	355	363	402			
R-squared	0.22	0.14	0.17			

- 1. Robust standard errors in parentheses are adjusted for clustering at the individual level.
- 2. * Significant at: * 10% level; ** 5% level; **
* 1% level

Table 8: OLS: Temporary Profit of Last-and-Final Bids

	Human Profit				Total	Profit
Treatment	iBEA	Vickrey	Vickrey (proxy)	iBEA	Vickrey	Vickrey (proxy)
SS_ℓ	3.04	2.72	3.13	16.29	10.85	13.17
SR_ℓ	3.03	1.80	2.77	13.61	8.56	10.95
RR_ℓ	3.93	3.30	4.27	15.05	9.58	11.71
SS_h	2.71	2.48	2.91	16.84	10.06	11.75
SR_h	3.53	2.96	3.67	15.77	9.75	11.87
RR_h	3.95	2.73	3.17	15.74	10.10	11.59
Human	7.26	4.51	5.03	21.79	13.52	15.08
		Rev	enue	Efficiency		
Treatment	iBEA	Vickrey	Vickrey (proxy)	iBEA	Vickrey	Vickrey (proxy)
SS_ℓ	13.87	20.12	16.58	78.0%	95.1%	83.5%
SR_ℓ	14.59	20.04	16.75	64.6%	72.7%	64.8%
RR_ℓ	12.93	18.88	15.84	65.6%	70.8%	61.8%
SS_h	13.05	20.44	17.33	79.9%	92.6%	78.9%
SR_h	12.77	19.92	16.62	71.7%	80.7%	69.1%
RR_h	11.92	17.81	15.21	62.7%	64.8%	55.0%
Human	5.30	16.42	13.51	56.2%	79.5%	67.2%

Table 9: Aggregate Performance of the Two Mechanisms: Session Average

	(1)	(2)	(3)	(4)			
Dependent Variable:	Human Profit	Total Profit	Revenue	Efficiency			
All Human-Robot Treatments							
Vickrey Proxy	-0.176	-3.514	2.915	-0.028			
	(0.386)	$(0.639)^{***}$	$(0.455)^{***}$	(0.028)			
D _H	-0.033	0.453	-0.594	-0.001			
	(0.313)	(0.529)	(0.402)	(0.022)			
# of Random Robots	0.441	-0.495	-0.617	-0.094			
	$(0.172)^{**}$	$(0.272)^*$	$(0.232)^{***}$	$(0.011)^{***}$			
Quiz Score	0.205	-0.31	0.449	0.019			
	(0.284)	(0.428)	(0.321)	(0.02)			
Constant	2.072	17.133	12.201	0.721			
	$(1.210)^{*}$	$(1.837)^{***}$	$(1.360)^{***}$	$(0.084)^{***}$			
Observations	1200	1200	1200	1200			
R-squared	0.01	0.07	0.07	0.07			
Only SS Treatments							
Vickrey Proxy	0.054	-3.536	2.843	0.007			
	(0.505)	$(0.882)^{***}$	$(0.787)^{***}$	(0.025)			
$D_{\rm H}$	-0.269	-0.485	0.025	-0.012			
	(0.415)	(0.748)	(0.716)	(0.019)			
Quiz Score	0.167	-1.032	1.192	0.028			
	(0.317)	$(0.605)^{*}$	$(0.513)^{**}$	(0.02)			
Constant	2.297	21.218	8.35	0.678			
	$(1.280)^*$	$(2.500)^{***}$	$(2.203)^{***}$	$(0.084)^{***}$			
Observations	400	400	400	400			
R-squared	0	0.10	0.09	0.01			

- 1. Robust standard errors in parentheses are adjusted for clustering at the individual level.
- 2. Significant at: * 10%; ** 5%; *** 1% level.

Table 10: OLS: Factors Affecting Aggregate Performance of the Two Mechanisms

Treatments	Human Profit	Total Profit	Revenue	Efficiency
(H_1)	$({\rm xHuman} > \cdots)$	$(\mathrm{xHuman} > \cdots)$	$(\mathrm{xHuman} < \cdots)$	(specified below)
i Human vs. iSS_l	0.016	0.066	0.016	0.016 (<)
i Human v s. iSR_l	0.027	0.016	0.016	$0.283~(\neq)$
i Human vs. iRR_l	0.027	0.026	0.016	$0.283~(\neq)$
i Human vs. iSS_h	0.016	0.098	0.016	0.016 (<)
i Human vs. iSR_h	0.016	0.043	0.016	$0.133~(\neq)$
i Human vs. i R h	0.027	0.043	0.016	$0.667~(\neq)$
iHuman vs. iRobot	0.012	0.028	0.008	$0.120~(\neq)$
v Human vs. vSS_l	0.026	0.043	0.020	0.016 (<)
v Human vs. \mathbf{vSR}_l	0.026	0.016	0.016	$0.667~(\neq)$
v Human vs. vRR_l	0.099	0.016	0.043	$0.133~(\neq)$
v Human vs. vSS_h	0.026	0.043	0.043	0.043 (<)
v Human vs. vSR_h	0.043	0.015	0.016	$0.667~(\neq)$
v Human vs. vRR_h	0.042	0.016	0.259	$0.053~(\neq)$
vHuman vs. vRobot	0.025	0.013	0.027	$0.937~(\neq)$

1. iHuman (vHuman) represents the human-human sessions in the *i*BEA (Vickrey) auctions.

2. iRobot (vRobot) represent the pooled human-robot sessions in the *i*BEA (Vickrey) auctions.

 Table 11: P-values of Aggregate Performance Comparisons between Human-Robot and
 All-Human Sessions



Figure 1: Simulated Profit for Human Bidder under Three Environments in the Vickrey Auction



Figure 2: Simulated Profit for Human Bidder under Three Environments in the *i*BEA



Figure 3: Proportion of MBR and non-MBR Bids in the iBEA Auction



Figure 4: Distribution of Observed Efficiency in the *i*BEA and Vickrey Auctions

References

- Ausubel, Lawrence M. and Paul R. Milgrom, "Ascending Auctions with Package Bidding," Frontiers of Theoretical Economics, 2002, 1 (1), 1-43. available at http://www.bepress.com/bejte/frontiers/vol1/iss1/art1.
- Banks, Jeffrey S., John O. Ledyard, and David P. Porter, "Allocating Uncertain and Unresponsive Resources: An Experimental Approach," *Rand Journal of Economics*, 1989, 20, 1–25.
- Brunner, Christoph, Jacob K. Goeree, Charles A. Holt, and John O. Ledyard, "Combinatorial Auctioneering," 2006. Caltech Working Paper.
- Bykowsky, Mark M., Robert J. Cull, and John O. Ledyard, "Mutually Destructive Bidding: The FCC Auction Problem," *Journal of Regulatory Economics*, 2000, 17 (3), 205–228.
- Cantillon, Estelle and Martin Pesendorfer, "Auctioning Bus Routes: The London Experience," in Peter Cramton, Yoav Shoham, and Richard Steinberg, eds., *Combinatorial Auctions*, Cambridge: MIT Press, 2006.
- Chen, Yan and Robert S. Gazzale, "When Does Learning in Games Generates Convergence to Nash Equilibrium? The Role of Supermodularity in an Experimental Setting," American Economic Review, December 2004, 94 (5), 1505–1535.
- ____ and Yuri Khoroshilov, "Learning under Limited Information," *Games and Economic Behavior*, July 2003, 44, 1–25.
- Clarke, Edward H., "Multipart Pricing of Public Goods," *Public Choice*, 1971, 11, 17–33.
- Compte, Olivier and Philippe Jehiel, "Auctions and information acquisition: sealed bid or dynamic formats?," *RAND Journal of Economics*, Summer 2007, *38* (2), 355–372.
- **Eisenberg, Anne**, "In Online Auctions of the Future, It'll Be Bot vs. Bot vs. Bot," *New York Times*, August 2000.
- Erev, Ido and Alvin E. Roth, "Predicting How People Play Games: Reinforcement Learning in Experimental Games with Unique, Mixed Strategy Equilibria," American Economic Review, September 1998, 88 (4), 848–881.
- Finholt, Thomas A., "Collaboratories," in B. Cronin, ed., Annual Review of Information Science and Technology, Vol. 36 American Society for Information Science

Washington, D.C. 2002, pp. 74–107.

- Fischbacher, Urs, "z-Tree: Zurich Toolbox for Ready-made Economic Experiments," Experimental Economics, 2007, 10 (2), 171–178.
- Goeree, Jacob K. and Charles A. Holt, "Hierarchical Package Bidding: A Paper & Pencil Combinatorial Auction," 2007. Caltech Working Paper.
- Groves, Theodore, "Incentives in Teams," Econometrica, 1973, 41 (4), 617–31.
- Isaac, R. Mark and Duncan James, "Robustness of the incentive compatible combinatorial auction," *Experimental Economics*, June 2000, 3 (1), 31–53.
- Kagel, John H., "Auctions: A Survey of Experimental Research," in J. Kagel and A. Roth, eds., *Handbook of Experimental Economics*, Princeton, New Jersey: Princeton University Press, 1995, pp. 131–148.
- **and Dan Levin**, "Behavior in Multi-Unit Demand Auctions: Experiments with Uniform Price and Dynamic Vickrey Auctions," *Econometrica*, March 2001, *69*, 413–454.
- Katok, Elena and Alvin E. Roth, "Auctions of Homogeneous Goods with Increasing Returns: Experimental Comparison of Alternative "Dutch Auctions," *Management Science*, August 2004, 50 (8), 1044–1063.
- Kwasnica, Anthony M., John O. Ledyard, David Porter, and Christine DeMartini, "A new and improved design for multi-object iterative auctions," *Management Science*, March 2005, 51 (3), 419–434.
- Ledyard, John O., David P. Porter, and Antonio Rangel, "Experiments Testing Multi-Object Allocation Mechanisms," *Journal of Economics and Management* Strategy, 1997, 6, 639–675.
- ____, Mark Olson, David P. Porter, Joseph A. Swanson, and David Torma, "The First Use of a Combined-Value Auction for Transportation Services," *Interfaces*, 2002, 32 (5), 4–12.
- Lucking-Reiley, David, "Auctions on the Internet: What's Being Auctioned, and How?," Journal of Industrial Economics, 2000, 48, 227–252.
- _____, "Vickrey Auctions in Practice: From Nineteenth-Century Philately to Twenty-First-Century E-Commerce," *Journal of Economic Perspectives*, 2000, 14 (3), 183–192.
- Milgrom, Paul R., Putting Auction Theory to Work, New York: Cambridge University

Press, 2004.

- Mishra, Debasis and David C. Parkes, "Ascending price Vickrey auctions for general valuations," *Journal of Economic Theory*, January 2007, *132* (1), 335–366.
- Ockenfels, Axel and Alvin E. Roth, "Last-Minute Bidding and the Rules for Ending Second-Price Auctions: Evidence from eBay and Amazon Auctions on the Internet," *American Economic Review*, September 2002, *92* (4), 1093–1103.
- Parkes, David C., "Auction design with costly preference elicitation," Journal Annals of Mathematics and Artificial Intelligence, July 2005, 44 (3), 269–302.
- **and Lyle H. Ungar**, "Iterative Combinatorial Auctions: Theory and Practice," in "Seventeenth National Conference on Artificial Intelligence (AAAI-00)" 2000, pp. 74–81.
- Rassenti, Stephen J., Vernon L. Smith, and Robert L. Bulfin, "A combinatorial auction mechanism for airport time slot allocation," *Bell Journal of Economics*, 1982, *XIII*, 402–417.
- Rothkopf, Michael H., Aleksandar Pekec, and Ronald M. Harstad, "Computationally Manageable Combinational Auctions," *Management Science*, August 1998, 44 (8), 1131–1147.
- Sandholm, Tuomas, David Levine, Michael Concordia, Paul Martyn, Rick Hughes, Jim Jacobs, and Dennis Begg, "Changing the Game in Strategic Sourcing at Procter & Gamble: Expressive Competition Enabled by Optimization," *Interfaces*, 2006, 36 (1), 55–68.
- Sarin, Rajiv and Farshid Vahid, "Payoff Assessments without Probabilities: A Simple Dynamic Model of Choice," Games and Economic Behavior, August 1999, 28 (2), 294–309.
- Takeuchi, Kan, John C. Lin, Yan Chen, and Thomas A. Finholt, "Scheduling with Package Auctions," 2008. University of Michigan Manuscript.
- Vickrey, William S., "Counterspeculation, Auctions and Competitive Sealed Tenders," Journal of Finance, 1961, 16, 8–37.
- Wulf, Wm A., "The collaboratory opportunity," Science, 1993, 261 (5123), 854–855.